Craig Chambers CSE 505 (Fall ’98) Reading Assignments

Reading Assignment #4

Text:

• chapter 5, Object-Oriented Programming: skim for concepts rather than syntactic details

“Using Types and Inheritance in OO Languages”:

read about different ways to use and misuse inheritance

“The Cecil Language: Specification and Rationale”:

concentrate on multimethods, fields via messages, & classless object model in section 2

“Self: The Power of Simplicity”:

a simplification of Smalltalk, using a purely OO prototype-based object model

“Virtual Classes: A Powerful Mechanism for Object-Oriented Programming”:

read about the utility of nested classes and virtual classes